Author's response to reviews

Title: A Young Patient with visual loss

Authors:

Mark Taubert (mtaubert@doctors.org.uk)
Tim C Dowd (TIMDREDAED@aol.com)
Angela Wood (Angela.Wood@stees.nhs.uk)

Version: 4 Date: 17 December 2007

Author's response to reviews:

Response to Reviewer:

Thanks again for going through our paper.
The reviewer writes:

"A well written and improved draft. I still think that one sentence in the abstract should be reconsidered:
"Results: Vitamin B12 and folate deficiency can increase the risk of thrombotic events by altering homocysteine metabolism."
This sentence is a) not supported by the findings in this case report and b) is not supported by any literature with which I am familiar.
Otherwise this is a well written, interesting, and unique case report."

This was an oversight on our part in the abstract and I have amended this as follows (track-changes used in text):

"Vitamin B12 and folate deficiency can lead to elevated levels of homocysteine. We found a large amount of published data relating central retinal vein occlusion to elevated homocysteine levels, but there was a lack of conclusive evidence for this association."

Thank you,

Mark Taubert